www.cambridge.org/pab

# Article

**Cite this article:** de la Parra, F., and R. Benson (2025). Diversification dynamics of vegetation during the Cenozoic in the Neotropics: a palynological perspective from Colombia. *Paleobiology*, 1–14. https://doi.org/10.1017/pab.2024.62

Received: 04 December 2023 Revised: 20 November 2024 Accepted: 02 December 2024

Handling Editor: Caroline Strömberg

**Corresponding author:** Felipe de la Parra; Email: felipe.delaparra@ecopetrol.com.co

© The Author(s), 2025. Published by Cambridge University Press on behalf of Paleontological Society. This is an Open Access article, distributed under the terms of the Creative Commons Attribution licence (http:// creativecommons.org/licenses/by/4.0), which permits unrestricted re-use, distribution and reproduction, provided the original article is properly cited.



# Diversification dynamics of vegetation during the Cenozoic in the Neotropics: a palynological perspective from Colombia

# Felipe de la Parra<sup>1,2</sup> D and Roger Benson<sup>1,3</sup>

<sup>1</sup>Department of Earth Sciences, University of Oxford, Oxford OX1 3AN, U.K. <sup>2</sup>Biostratigraphy Team, Instituto Colombiano del Petróleo-ECOPETROL S.A., 681011 Santander, Colombia <sup>3</sup>Richard Gilder Graduate School and Division of Paleontology, American Museum of Natural History, New York 10024, U.S.A.

## Abstract

The Neotropics host the highest diversity of plants on the Earth today and have done since at least the late Paleogene (~58 Ma). Several mechanisms have been proposed to explain this elevated diversity, but the empirical patterns of Neotropical plant diversification that would test key aspects of those mechanisms are still unclear. We use an extensive palynological database from northern South America to characterize patterns of extinction, origination, and diversity and their possible drivers since the Paleogene. The foreland Llanos basin of Colombia preserves the evolutionary history of Neotropical vegetation as well as the geological evolution of northern South America, offering a unique opportunity to study the relationship between the geological and fossil records. The palynological record of the Llanos basin has been intensely studied mainly for oil exploration, and we use this information to infer the evolutionary history of Neotropical vegetation in Colombia during the Cenozoic. There is no straightforward relationship between global temperature and Neotropical plant diversity. Nevertheless, environmental change had an important influence on the dynamics of diversification, especially during volatile climate intervals such as the Paleocene-Eocene and the Pleistocene. Pulses of regional extinction were driven by large-scale temperature excursions, including both warming and cooling phases. Time-lagged origination pulses results in rapid floral replacement on a timescale of 1 Myr. Origination and extinction are essentially balanced on long timescales, leading to a near-zero long-term net diversification rate. Regional geological events, like the uplift of the Andean Cordillera, and changes in paleogeography also played an important role in Neotropical plant diversification.

## **Non-technical Summary**

The Neotropics maintain the highest biodiversity of the planet, exert control in global climate, and play an important role in the dynamics of our planet. Researches from a wide range of disciplines are very interested in understanding how diversity in the Neotropics was produced and how it is maintained through deep time. Despite important advances in our understanding of the origin and diversification of Neotropical biota, we still do not have a clear idea about the patterns of diversification through geological time. This presents a significant knowledge gap that needs to be filled. Fossils are necessary to understand the history of the Neotropical biota. From the perspective of plants, palynology represents the best proxy to study the macroevolution of vegetation. Pollen and spores fossilize readily, they are produced in large amounts and are found in a wide variety of sedimentary environments, and they have high stratigraphic resolution. In this work, we used a compilation of palynological information from Colombia to explore patterns of plant diversification during the Cenozoic and their possible drivers. Changes in global temperature had an important influence on the dynamics of diversification, especially during volatile climate intervals. Paleogeographic changes and regional geological events also had an important role in modeling diversity. Origination and extinction are balanced on long timescales, with pulses of extinction followed by rapid floral replacement on a timescale of 1 Myr. The macroevolutionary history of Neotropical vegetation is a complex puzzle that requires the integration of data from multiple areas of research in biology and earth-physical sciences. The purpose of our work is to provide a small piece of evidence that helps to reconstruct this interesting puzzle.

## Introduction

The Neotropics, the geographic region that corresponds to the tropics of the New World (Morrone 2014), host tremendous extant plant diversity and are therefore a model system for understanding the origin and maintenance of biodiversity (Gentry 1992; Perez-Escobar et al.



2022). Neotropical plant diversity has been attributed to various mechanisms involving both biotic and abiotic factors, including the occurrence of a relatively stable long-term climate regime compared with northern temperate regions, recent diversification events during glacial-interglacial cycles (Haffer 1969), and high origination rates or low extinction rates in the tropics (Stebbins 1974). The Stebbins' (1974) hypothesis poses that the tropics have acted as a "cradle," a "museum," or both, with taxa preferentially originating in the tropics before expanding into higher latitudes (Jablonski et al. 2006), a mechanism that has found strong support from phylogenetic studies of modern flowering plants (angiosperms), which comprise at least 80% of living plant species. Phylogenetic studies of extant angiosperms suggest that per-species origination and extinction rates in the tropics are not substantially different from those outside the tropics. The Neotropics are therefore distinctive among global floristic regions as a macroevolutionary source region that has exported species to other regions thorough most of the Cenozoic (Antonelli et al. 2015).

Although climate is one of the best predictors of species diversity today, its role in producing and maintaining plant diversity in deep time is still under debate. Palynological analysis from the Paleogene of Colombia indicates a correlation between fluctuations in diversity and changes in global temperature (Jaramillo et al. 2006), suggesting that climate variation may be a direct driver of Neotropical plant diversity, possibly by driving variation in available area (Jaramillo et al. 2006). Global warming episodes, such as the Paleocene-Eocene thermal maximum (PETM) and the mid-Miocene climatic optimum (MMCO) are also characterized by short-term increases in origination rates, causing increases in diversity (Jaramillo et al. 2010; D'Apolito 2016). Evidence so far, therefore, suggests that transient climatic warming events promote diversification. Nevertheless, a correlation between diversity and temperature is not very evident on longer timescales. For example, changes in palynological diversity during the Neogene are not closely related to temperature variation (Hoorn and Wesselingh 2010).

A tectonic-related explanation of Neotropical plant diversification has also been proposed (Gentry 1982). The uplift of the northern Andean Cordillera, of which the most intense pulse occurred during the Miocene-Pliocene (Gregory-Wodzicki 2000; Hoorn et al. 2010; Siravo et al. 2019) resulted in substantial changes that affected the paleogeography of northern South America and especially the Amazon Basin (Mora et al. 2010; Rodazz et al. 2010; Leite et al. 2016; Perez-Escobar et al. 2022). This orogenic pulse promoted habitat heterogeneity, created biotic corridors, intensified the nutrient supply, and changed the hydrological and climate regime of the entire continent (Mora et al. 2008; Hoorn and Wesselingh 2010 and references therein). These changes could favor adaptative radiations, produce geographic vicariance, and promote allopatric speciation that may have increased the plant diversity of the region. However, the relationship between orogenic activity of the Andes and long-term patterns of plant diversity has been difficult to prove. The palynological diversity curve proposed by Jaramillo et al. (2006), which is the most complete to date, does not include the Miocene and is restricted to the interval between 66 and 20 Ma. Moreover, it was constructed using rangethrough methods that are highly prone to distortions such as edge effects (Foote 2000), and more recent methods of diversity estimation (e.g., Alroy 2010a, 2014; Chao and Jost 2012; Chao et al. 2014; Close et al. 2018) have not yet been applied. Other palynological studies in the Amazon basin of Brazil have been restricted to short intervals of the Miocene (Silva-Caminha et al. 2009; D'Apolito 2016; Leite et al. 2016).

Fossils provide unique observations of species in deep time that can shed light on the mechanisms of diversification and are an especially powerful tool for characterizing variation in diversity and rates of diversification and turnover through deep time (Alroy 2000; Foote 2000; Foote and Miller 2006; Silvestro et al. 2019; Flannery-Sutherland et al. 2022). Diversity estimates and measures of taxonomic rates (origination and extinction) have been widely used to study the evolutionary history of organisms, including marine invertebrates (Sepkoski 1996; Foote 1988; Alroy 2008); mammals (Alroy 2009); marine microorganism (Lloyd et al. 2011; Lazarus et al. 2014; Fraass et al. 2015); and tetrapods (Benton 1995; Benson et al. 2016). Although the counts of fossils and the methods used for macroevolutionary estimates can be distorted by factors like sampling bias, the Pull of the Recent, and the Signor-Lipps effect (Raup 1972; Foote and Miller 2006), there is substantial evidence that for a broad perspective on global and regional diversity trends, the methods and compendia of fossils have now matured sufficiently to capture the main biotic transitions preserved in the geological record (Foote and Miller 2006; Alroy 2010a; Quental and Marshal 2010). As a consequence, several approaches and methods to estimate diversity and taxonomic rates from the fossil record are now available (Alroy 2000, 2014; Foote 2000; Foote and Miller 2006; Silvestro et al. 2018, 2019).

The macrofossil record of plants has been successfully used to test macroevolutionary hypotheses. Examples include the origin and diversification of vascular plants (Silvestro et al. 2015), the dispersal and extinction rates across continents during the Cenozoic (Silvestro et al. 2016), the role of biotic and abiotic mechanisms in controlling species diversification in ferns (Lehtonen et al. 2017), and the origin of the modern Neotropical forest (Carvalho et al. 2021).

Microfossils provide another source of information for vegetation dynamics in the geological past, specifically pollen and spores. Due to their role in reproduction, plants tend to produce large amounts of pollen and spores that fossilize readily (Traverse 2007). It has been argued that because pollen and spores lack a sufficient amount of diagnostic synapomorphies that prevent accurate taxonomic classification, they are not a reliable source of information in evolutionary studies of plants (Silvestro et al. 2015). However, it has also been demonstrated that diversity in pollen rain follows the trend of vegetation diversity over ecological gradients (Weng et al. 2006; Birks et al. 2016; Felde et al. 2016. Gosling et al. 2017) and reflects local to regional vegetation conditions well (Gosling et al. 2009; Jantz et al. 2013; Matthias et al. 2015; Julier et al. 2017).

The fossil record of Neotropical macro-plant remains (e.g., leaves, wood, flowers, fruits) is deficient and poorly studied (Wing et al. 2009). Neotropical rock outcrops are scarce and difficult to access due to dense living vegetation. This condition has prevented a complete and continuous fossil record of this region. However, and despite the difficulty of obtaining a good and complete paleobotanical record from the Neotropics, there is one region in Colombia that, due to its importance for oil exploration, has been intensely studied in terms of geology and palynology. This region corresponds to the Llanos basin, a foreland basin that is limited by the Guiana Shield and the Amazon basin to the east and the eastern Andean Cordillera to the west. Due to its geographic position, the Llanos basin preserves a very complete record of the geological and palynological history of the Neotropics (Fig. 1). This basin can therefore be considered to be a natural laboratory to study the relationship between the uplift of the northern Andean Cordillera with its climatic and environmental changes and the macroevolutionary history of Neotropical vegetation during the Cenozoic.

The Llanos basin has been intensely studied in terms of palynology, and an abundant record of fossil pollen and spores from



Figure 1. Map of Colombia showing the geographic location of the sections (wells and outcrops) used in this study. The red circles correspond to the sections used for our estimations of diversity and taxonomic rates. WC, Western Andean Cordillera; CC, Central Andean Cordillera; EC, Eastern Andean Cordillera; CAT, Catatumbo basin.

subsurface drilling/coring can therefore be used to provide insight into the deep-time history of Neotropical plants. In spite of palynology's great potential, only a few studies have used it to approach the evolutionary dynamics of Neotropical vegetation (Hoorn 1993; Jaramillo and Dilcher 2001; Jaramillo et al. 2006; D'Apolito 2016; Leite et al. 2016; Jaramillo 2019). So far, these studies have focused on local geographic areas and were restricted to specific and relatively short intervals of the Cenozoic. This has given us only snapshots of the evolution of Neotropical vegetation and prevented us from attaining a larger understanding of the drivers of Neotropical plant diversity.

Here, we use a high-resolution palynological record obtained from 6098 samples, representing 110 sections (including wells and outcrops) drilled mainly in the Llanos basin of Colombia. We combine the samples to produce a composite section spanning from 66 Ma to the present, estimating diversity and taxonomic rates (origination and extinction) using sampling-standardized methods and evaluating their possible drivers.

## **Materials and Methods**

#### **Composite Section**

We analyzed a palynological inventory of 110 sections (Fig. 1; Supplementary Material 1), including both wells and outcrops. Of these, 63 correspond to sites previously analyzed by Jaramillo et al. (2011). Our analysis includes a total of 6098 samples and 2777 morphospecies, of which 1200 morphospecies have informal names only. Nevertheless, we are confident about the consistency of morphospecies delimitation, which results from a detailed morphological and taxonomic comparison with species stored in a database hosted by the Biostratigraphy Team of the Instituto Colombiano del Petróleo (ICP-Ecopetrol S.A.), including photomicrographs and morphological descriptions of thousands of palynomorphs found in the Neotropics. All the palynological slides used in this study are stored in the palynological collection of the Ecopetrol Core Library, Instituto Colombiano del Petroleo (ICP), Santander, Colombia.

Samples from all sections were combined in a single composite section using graphic correlation (Shaw 1964; Edwards 1984; Supplementary Material 2). The final composite section includes sediments that accumulated in fluvial to coastal plain environments between the Paleocene and the Holocene (66–0 Ma) (Cooper et al. 1995; Bayona et al. 2007; Parra et al. 2009; Bande et al. 2012; Reyes-Harker et al. 2015). Each part of the composite section includes information from several sections and therefore from different depositional environments and biomes. This approach reduces the chances that differences in environment of deposition influence changes in taxonomic diversity over time. The age of each sample was calculated using the equations of the lines of correlation and the calibration points used by Jaramillo et al. (2011), which include foraminiferal biostratigraphy, stable carbon isotope stratigraphy ( $\delta^{13}$ C), and key stratigraphic data.

Because most of the material corresponds to ditch-cutting samples, borehole caving can artificially increase the first appearance datum (FAD) of the species, making the apparent time of origination of the species older than the actual time of origination. This could act as a systematic bias in macroevolutionary estimates. To identify the potential influence of caving on the ranges of the species, and therefore on the diversity and taxonomic rates estimates, two versions of the composite section were used. One version (complete version) included all occurrences of all 2777 morphospecies, regardless of the type of samples from which their FADs were interpreted. The other version (restricted version) included only those morphospecies (193) whose FADs have been confirmed using core or outcrop samples. If caving did not substantively affect the ranges of the species, we would expect to find similar results with both datasets, which we do.

#### Diversity

We used two methods to estimate diversity. First, we estimated standing diversity from our dataset using first appearance data (FADs) and last appearance data (LADs), using a range-through assumption (i.e., assuming that each morphospecies was present at all times through its first and last appearance) (Foote and Miller 2006). This is the approach used by the previous study of large-scale diversity patterns of Neotropical palynomorphs (Jaramillo et al. 2006), but suffers from substantial edge effects (Foote 2000). Indeed, our uncensored, range-through diversity curve and that of Jaramillo et al. (2006) both have a characteristic long-term hump shape consistent with a strong influence of edge effects (Signor and Lipps 1982). Therefore, we also applied a censored version of range-through diversity that makes use of only those species with sampled ranges below a threshold duration. Censorship of the data to include only short-ranged taxa should limit the duration of potential edge effects, rendering the central part of the time series more interpretable and indicates a substantially different diversity pattern (Supplementary Material 3, Supplementary File 1).

Accurate diversity estimates should statistically remove biases imposed by both the counting method and variation in sampling intensity through time. Variation in the quantity of sampling is problematic, because estimated differences in diversity between two samples can be explained by differences in sampling effort and not biological factors. The goal of most resampling methods is to obtain an empirical approximation of a sampling distribution of a statistic of interest and can be used to standardize samples, test statistical hypothesis, estimate standard errors, and carry out other statistical evaluations of empirical datasets (Kowaleski and Novack-Gottshall 2010). In the specific case of diversity estimation, randomized subsampling protocols seek to hold each time interval to a quota of data items. This quota can be expressed in terms of (1) number of taxonomic occurrences; (2) number of fossil collections; and (3) occurrences represented by the collections (Alroy 2010a). Most of the methods to estimate diversity, including classic rarefaction, subsample the data from each interval to attain an entirely uniform and fixed quota of data items (i.e., counts of specimens, counts of occurrences, or counts of fossils collections) (Alroy et al. 2001; Bush et al. 2004). However, uniform-sampling schemes tend to flatten out variation in diversity through time because they are designed to count items, ignoring systematically two prominent features of the fossil record, rarity and commonness (Alroy 2010b). Uniform subsampling is therefore different from fair or representative subsampling. As pointed out by Alroy (2010b), when the actual richness of the species pool is low, it is not necessary to sample very hard, but when richness is high and many taxa are rare, it may be necessary to subsample more intensively to achieve comparable coverage of the species pool. In this case, uniform-sampling schemes fail by undersampling the richer species pool.

Shareholder quorum subsampling (SQS [Alroy 2010a] or "coverage-based rarefaction" [Chao and Jost 2012]) use sampling standardization to estimate richness by subsampling to equal coverage of the species occurrence-frequency distribution. This is intended to achieve fair rather than uniform sample frequencies by tracking not the number of items that are drawn but the coverage of the underlying species pool represented by the species that have been drawn. Coverage is the proportion of the entire frequency distribution represented by the species found so far. The coverage of observed data is modified to estimate the coverage of the real taxon distribution for each sample pool (Alroy 2010a,b). This is achieved by multiplying the coverage of the observed data by Good's u; the proportion of occurrences representing non-singleton taxa, which is a measure of sample completeness (Alroy 2010a). Standardization in SQS is achieved therefore by drawing enough collections in each time interval to generate the same coverage level. The quorum (q) refers to a certain amount of coverage, if q = 0.30 sampling stops when 30% of the distribution is covered. Alroy (2010a,b) proposed that a quorum of 0.4 is the minimum required to recover consistent relative estimates of diversity.

SQS was implemented in three ways using the R package iNext v. 2.0.20 (Hsieh et al. 2016). First, we applied SQS subsampling within individual samples, using the species abundance-frequency distribution (Supplementary File 2). Only samples with counts greater than 50 palynomorphs were included in this analysis as a quality-control threshold. Second, we applied SQS using the pooled species abundance-frequency distributions of all samples within 1 Myr intervals (i.e., using the sum of the abundances reported in all the samples where each species was found) (Supplementary File 3). Third, we subsampled from the species occurrence-frequency distributions, using presence/absence data for species within samples (Supplementary File 4). In all the approaches, SQS diversity was estimated using a quorum level of 0.6.

The changes in palynological composition were assessed using a constrained hierarchical cluster (Birks and Gordon 1985). The cluster was constructed from the pooled species presence/absence data of all samples within 1 Myr intervals using the range-through assumption after removing singletons. Because each part of the composite section incorporates information from sections deposited in different environments of deposition and biomes, it is difficult to separate the contribution of each biome to the diversity curve. This approach, and the lack of knowledge about the botanical affinity of most of the palynomorphs, prevents an accurate and reliable reconstruction of the geographic distribution of biomes during each time interval. However, the cluster analysis allows us to establish the main differences in the overall plant composition throughout the Cenozoic. For the cluster analysis, we used the package rioja (Juggins 2023; R Core Team 2024).

#### **Taxonomic Rates**

By taxonomic rates, we mean the rates at which new morphospecies originate (origination) and existing species become extinct (extinction) (Foote and Miller 2006). A practical impact of estimating origination and extinction rates considering entire age ranges (the range-through assumption) (Foote 2000) is that those metrics introduces a series of biases, such as the Signor-Lipps effect (Signor and Lipps 1982), the Pull of the Recent, and other edge effects, which are difficult to quantify and remove. To correct some of these problems, Alroy (2008, 2010a, 2014) introduced a series of equations (the three-timer equation and the gap-filler equation) to estimate origination and extinction that do not make use of the range-through assumption and instead focus on occurrence data that indicate which taxa are sampled in which time interval. These equations involve examining only a limited sampling window consisting of four consecutive time intervals (*i*: focal bin; i - 1, i + 1, i + 2). Similar to the approach proposed by Foote (2000), Alroy's method counts the number of taxa sampled in each one of four different categories: two-timers (2T): taxa sampled immediately before (i - 1) and within focal bin (i); three-timers (3T): taxa sampled in the focal bin (i) and immediately before (i - 1) and after (i + 1); part-timers (p): taxa sampled before and after but not in the focal bin; gap-fillers (g): taxa sampled in intervals i - 1 and i + 2 but not within interval i + 1 or in interval i + 2 and i + 1, and that may or may not be found in the focal interval. The gap-filler equations work with the four interval-moving windows and consider the categories 2T, 3T, p, and g. Although the gap-filler origination and extinction rate equations use some of the same counts, there is no spurious correlation between them (Alroy 2014). Origination and extinction rates were calculated using the modified versions of the gap-filler equations proposed by Alroy (2015; modified from Alroy 2014) implemented using custom code. Gap-filler rate estimates incorporate an estimate of sampling rate, allowing estimation from incompletely sampled species occurrence data.

Simulations suggest that all methods to estimate taxonomic rates (including boundary-crosser and three-timer and gap-filler equations) might yield different values depending on whether the data are sampling standardized (Alroy 1996; Alroy 2010b). Gapfillers might produce less noisy values if sampling probabilities are made more uniform. The most employed standardization method is classic rarefaction of taxonomic occurrences (Foote and Miller 2006) and SQS (Alroy 2010a). Because the three-timer and gapfiller methods build in assumptions about the equality of sampling probabilities from one interval to the next, they are better met by data that are previously standardized using classic rarefaction (Alroy 2014). In our work, the taxonomic rates using the gap-filler method are based on classic rarified data.

We divided the complete composite section into 1 Myr time intervals (bins) and standardized by randomly drawing entire collections up to a uniform quota of 85 samples per bin. Resampling was repeated 1000 times, and origination and extinction were calculated for each one of the iterations. The median origination and extinction rates were calculated from 1000 subsampling iterations (Supplementary Files 5, 6). We calculated diversification rates as being origination rate minus the extinction rate. We calculated turnover rates as the smallest value of either origination or extinction rate within each time interval. This is an index of the proportion of species that became extinct and were quantitatively replaced by new species within each interval.

#### Results

#### Diversity

The analyses of the two versions of the composite section yield similar diversity estimates using equal coverage subsampling (SQS; Alroy 2010a; Chao and Jost 2012; Supplementary Material 4, -Supplementary Fig. 2). This indicates that any potential effect of borehole caving on diversity estimation is almost undetectable. Our interpretation therefore focuses on the complete composite section, which incorporates information from a much greater number of species (2777 species, compared with 193 in the restricted composite section). The abundance-based, per-sample analysis yields a cloud of points in which each point represents one sample (Supplementary Material 5, Supplementary Fig. 1). Variation in sampling intensity through the Cenozoic is clearly evident, with relatively sparse sampling during the early-middle Eocene and late Miocene-Pleistocene. This analysis also demonstrates substantial variation in diversity among samples of the same interval through the Cenozoic (Supplementary Material 5).

Diversity curves based on SQS of occurrence data agree with the values obtained from abundance data (Supplementary Mate rial 5, Supplementary Fig. 3) regarding the most prominent patterns. However, in the occurrence version, the sampled coverage of the underlying taxon pool is insufficient for several bins of the Eocene and Miocene and does not provide information of diversity. We used a quorum level of 0.6, because at this quota it was possible to obtain SQS diversity estimates for all the bins. Is important to note that different quorum levels reproduce very well the most prominent changes in diversity (Supplementary Material 5, Supplementary Fig. 4).

Our curve exhibits three broad phases of diversity (Fig. 2). Diversity during the early Cenozoic (Paleocene–Eocene) shows high-amplitude short-term fluctuations, whereas later Cenozoic diversity estimates (Oligocene–Miocene) were more stable on short timescales, followed by a return to fluctuating diversity estimates in the past 5 Myr (Pliocene–Recent) (Fig. 2). SQS diversity does not correlate with  $\delta^{18}$ O using the complete dataset (66–0 Ma) or when they are analyzed separately (66–34.5 Ma and 33.5–0.5 Ma) (Table 1). The overall shape of our curve differs markedly from a previous standing diversity curve obtained using range-through taxon counts (Jaramillo et al. 2006). We attempted to mitigate these edge effects by limiting analysis to only those taxa with relatively short ranges, spanning less than 70 Myr, 20 Myr, and less than 10 Myr. The resulting diversity curves (Supplementary Material 3)



**Figure 2.** Changes in palynological diversity and taxonomic rates during the Cenozoic. **A**, Cenozoic oxygen isotope record for benthic foraminifera showing the most prominent global climate change events during the Cenozoic (modified from Westerhold et al. 2020). **B**, Pollen and spore diversity estimated using abundance-based shareholder quorum subsampling (SQS) at a quorum (*q*) of 0.6. **C**, Extinction rates estimated using the gap-filler equation (Alroy 2010a, 2014). Extinction shows an increasing trend during the Paleocene, short-amplitude fluctuations during the Eocene to middle late Miocene interval, and an increasing trend during the late Miocene to Pleistocene. **D**, Origination rates estimated using the gap-filler equation (Alroy 2010a, 2014). Origination exhibits wide amplitude fluctuations during the Paleocene to middle Eocene, with at least three outstanding peaks occurring during the early Paleocene and early and middle Eocene. From the middle Eocene to the late Miocene, the origination rates exhibit short-amplitude fluctuations without apparent peaks. The Pleistocene is characterized by an abrupt increase in origination. The shaded area in extinction and origination rates indicates 95% confidence intervals derived from iterations of the sampling-standardization procedure.

Start interval	End interval	Treatment	N (intervals)	Coefficient (occurrences)	<i>p</i> -value	Coefficient (abundances)	<i>p</i> -value
Paleocene	Recent		66	-0.347	0.005*	-0.283	0.022*
Paleocene	Recent	Detrended	65	0.036	0.778	0.148	0.241
Paleocene	Eocene	Greenhouse	32	0.026	0.886	0.181	0.319
Oligocene	Pleistocene	Icehouse	29	0.19	0.322	0.126	0.525

**Table 1.** Spearman's rank correlations between shareholder quorum subsampling (SQS) diversity (quorum 0.6) and the  $\delta^{18}$ O global climatic proxy. Detrending was carried out using first differences.

\*Significant.

mimic some, but not all, features of our SQS diversity curves, including high diversity during the mid-late Eocene and relatively low, stable diversity since then.

Diversity increased during the Paleocene-early Eocene, from initially low levels associated to the effects of the K-Pg boundary extinction event. Short-lived peaks of high diversity approximately coincide with global warming events of the PETM and the early Eocene climatic optimum (EECO) (Fig. 2). Low diversity values during the middle Eocene were followed by a peak coinciding with the mid-Eocene climatic optimum (MECO) at ~42.5 Ma. Diversity then decreased until the end of the Eocene and remained low during the Oligocene and early Miocene (Fig. 2). The early Oligocene Oi-1 glaciation (~33.7 Ma) coincided with low diversity, whereas the late Oligocene Mi-1 glaciation (~24 Ma) coincided with transient, higher diversity before diversity decreased during the final 3 Myr of the Oligocene. A pattern of increasing diversity through the Miocene culminated in relatively high diversity after the MMCO (15-17 Ma) before decreasing to lower levels in the late Miocene (Fig. 2). Diversity was relatively high during the Pliocene and Pleistocene, with peaks during the early Pliocene and early Pleistocene (Fig. 2).

The constrained stratigraphic cluster shows two main groups, Paleocene–Eocene cluster (PE) and Oligocene–Miocene–Plio-Pleistocene cluster (OMP), which in turn are divided into smaller groups (Fig. 3). The PE and OMP clusters separate two different palynofloras. The PE cluster includes samples from the Paleocene and the Eocene, while the OMP includes samples from the Oligocene, Miocene, Pliocene, and Pleistocene (Fig. 3). The PE cluster is subdivided into two groups. The first group includes samples exclusively from the Paleocene, while the second group divides the Eocene into two groups. The first group includes samples from the Oligocene and the early-middle Miocene, while the second group encompasses samples from the late Miocene, Pliocene, and Pleistocene (Fig. 3).

#### Taxonomic Rates

Origination and extinction rates show a pattern of high volatility during the Paleocene–Eocene, with variation in extinction rates being especially high during the Paleocene, and variation in origination being remaining high throughout the Paleocene–Eocene. This was followed by stability at low values for most of the Oligocene–Miocene, and an increase to higher volatility during the Pliocene–Pleistocene (Fig. 2). Variation in extinction and origination rates is broadly congruent with patterns of relatively volatility in the SQS diversity curve, although it does not result from estimation from the same underlying data: origination and extinction rates were estimated using a subsample moving window method, based only on the patterns of presence/absence of taxa within four



Figure 3. Constrained stratigraphic cluster showing differences in plant composition through the Cenozoic. Two main groups, Paleocene–Eocene cluster (PE) and Oligo-cene–Miocene–Plio-Pleistocene cluster (OMP), separate the Paleocene and Eocene from the Oligocene, Miocene, Pliocene, and Pleistocene.



Figure 4. Extinction vs. origination rates. Correlation is maximized at a time lag of one interval (Table 2). A, Extinction rate lags origination rate; B, extinction rate vs. origination rate; C, extinction preceded origination.

consecutive time bins (Alroy 2015), whereas SQS diversity is estimated only from patterns of relative abundance within each time bin (Alroy 2010a; Chao and Jost 2012). Peaks of elevated extinction are evident during the early Paleocene (64-63 and 62-61 Ma), the Paleocene/Eocene transition (57-56 Ma), and during the Pliocene (5-2 Ma), with smaller peaks evident at other time intervals (e.g., 53-52 Ma, 50-49 Ma, 42-41 Ma), including the Eocene/Oligocene transition (34-33Ma) (Fig. 2). Peaks of elevated origination are evident during the early Paleocene (63-62 Ma), the EECO (53-52 Ma), MECO (43-42 Ma), Pliocene and Pleistocene (5-0 Ma), with smaller peaks frequently evident in the Paleocene and early to mid-Eocene, as well as during the Eocene/Oligocene transition (34-33 Ma) and middle Miocene (Fig. 2). The correlation between extinction and origination rates is maximized at a time lag of one interval (Fig. 4, Table 2), indicating that regional extinctions are generally followed by origination, with a delay of approximately 1 Myr (Table 2). Comparison between origination and extinction for a time lag of zero is significant in only 9.3% of iterations of our sampling-standardization procedure (Table 2).

Diversification rates also show their highest volatility in the Paleocene–middle Eocene and Pliocene–Pleistocene, with especially high relatively stability during the Oligocene (Fig. 5). A period of high-amplitude fluctuations of both diversification and turnover rates is concentrated in the Paleocene and Eocene, marked by high positive diversification rates at 63–62 Ma and 43–42 Ma, and high negative diversification rates at 62–61 Ma and 57–56 Ma, immediately before the Paleocene/Eocene boundary and during the Pliocene (5–2 Ma). Large positive diversification shifts do not generally coincide with high turnover. However, large negative diversification shifts often do, as seen around the Paleocene/Eocene boundary and during the early-late Pliocene transition. High turnover also occurs at 53–52 Ma, around the onset of the EECO and associated with only a weak positive diversification (Fig. 5).

Diversification and turnover show low to null variability during the Oligocene and Miocene, interrupted by relatively higher levels of turnover at the Eocene/Oligocene boundary and in the middle Miocene, possibly related to the MMCO (Fig. 5). A noticeable difference occurs in the Oligocene, where the turnover rate exhibits low-amplitude fluctuations, while the diversification rate is approximately zero during this interval (Fig. 5).

#### Taxonomic Rates versus Climate

We compare our diversity and taxonomic rate estimates with the climatic curve proposed by Westerhold et al. (2020), which indexes global climate rather than regional temperature variation. However, there is no obvious reason why this disconnect of scale could cause false-positive correlations, and it therefore provides a conservative test of the potential effects of temperature variation on Neotropical plant diversification. Temperature change from one interval to the next is not correlated with either origination or extinction rates. However, stronger correlations are obtained when considering the positivized net temperature change over longer intervals, maximizing at 3 Ma, for which the correlation with extinction rates is positive and significant (Table 3, Fig. 6). This effect is especially evident during the middle Paleocene cooling (up to 61 Ma), warming up to the Paleocene-Eocene boundary (56 Ma), and Pleistocene cooling (up to 3–2 Ma) (Fig. 6). A less prominent effect is also evident at the Eocene/Oligocene boundary (33.9 Ma).

### Discussion

Our diversity curve reveals differential dynamics of climate and vegetation during greenhouse and icehouse intervals. Greenhouse

**Table 2.** Summary of Spearman's rank-correlation test results betweenorigination rates and extinction rates at a lag time. Results are summarizedacross 1000 iterations subsampling to a uniform quota of 85 occurrences withineach 1 Myr interval

Extinction bin	Origination bin	Correlation coefficient (median)	<i>p</i> -value (mean)	Proportion significant (p < 0.05)
i – 1	i	-0.142	0.261	0.192
i – 1	i	0.287	0.050*	0.635
i	i	0.05	0.44	0.093
i + 1	i	0.01	0.498	0.055
i + 2	i	0.191	0.144	0

\*Significant.



Figure 5. Changes in turnover and diversification rates during the Cenozoic. **A**, Turnover rates (origination rate + extinction rate) show high volatility during the Paleocene to early Eocene. **B**, Diversification rates (origination rate – extinction rate) exhibit high volatility in the Paleocene–middle Eocene and Pliocene–Pleistocene intervals and stability during the Oligocene. The shaded area indicates 95% confidence intervals derived from iterations of our sampling-standardization procedure.

conditions of the Paleocene–Eocene were globally warmer than today, with minimal or no glaciation and lacking polar ice caps (Huber and Thomas 2009), and show volatile patterns of diversity and diversification (Fig. 2). In contrast, icehouse conditions commenced during the middle Eocene to early Oligocene with global cooling and the onset of Southern Hemisphere glaciation and expanded during the late Miocene–early Pliocene with further cooling and the onset of Northern Hemisphere glaciation. This long interval showed relatively greater climate stability, and we find also more stable patterns of diversity and diversification (Figs. 2, 5). The cluster analysis also reveals significant differences in terms of plant composition between the greenhouse and icehouse intervals. The two most relevant groups (PE and OMP) clearly separate samples of the Paleocene and Eocene from samples of the Oligocene, Miocene, Pliocene, and Pleistocene (Fig. 3). The change from greenhouse to icehouse conditions was accompanied by a change in

Temperature change interval (Ma)	Directional or positivized	Correlation coefficient (extinction)	<i>p</i> -value	Correlation coefficient (origination)	<i>p</i> -value
1	Directional	0.04	0.754	0.02	0.878
2	Directional	0.234	0.067	-0.036	0.78
3	Directional	0.178	0.169	0.016	0.902
4	Directional	0.128	0.327	-0.045	0.73
1	Positivized	0.013	0.921	-0.011	0.932
2	Positivized	0.233	0.068	0.009	0.942
3	Positivized	0.272	0.034*	0.22	0.083
4	Positivized	0.182	0.164	0.196	0.129

Table 3. Spearman's rank-correlation test results between net temperature change over multi-million-year intervals and taxonomic rates at a lag time.

\*Significant.



Figure 6 Temperature vs. extinction rates. A, Correlation between temperature change between one interval to the next and extinction rates estimated using the gap-filler equation (Alroy 2010a, 2014); B, strong, positive, and significant correlation between extinction rate and temperature change over long intervals of time (e.g., 3 Myr).

plant composition that is reflected in the diversity and taxonomic estimates.

The diversity curve also reveals an early increasing trend during the Paleocene–Eocene greenhouse interval, culminating in high diversity by the early Eocene. Regional extinction and origination rates are elevated and highly fluctuating during this interval (Fig. 2). The relatively low palynological diversity and high volatility of the extinction and origination rates during the Paleocene may be a direct consequence of the effect that the K-Pg transition had on tropical vegetation, potentially reflecting long-term ecosystem instability in the wake of this mass extinction event. This resulted in dramatic differences in palynological composition between the Cretaceous and the Paleocene (de la Parra et al. 2022). Plant extinctions levels over the K-Pg boundary were high in the Neotropics, causing low diversity levels for at least 6 Myr after the extinction event (Carvalho et al. 2021), as reflected in our diversity curve (Fig. 2).

The increasing trend in diversity is punctuated by two outstanding peaks that show a good correspondence with global warming events of the PETM and the EECO (Fig. 2). The PETM is recognized as a rapid (<20 ka) worldwide warming event that was possibly driven by the destabilization of carbon from surface sedimentary reservoirs such as methane hydrates (Dickens et al. 1995) or by the volcanisms associated with the North Atlantic Igneous Province (Gutjahr et al. 2017). In northern South America, palynological studies reveal a rapid and distinct increase in plant diversity and origination rates through the PETM (Rull 1999; Jaramillo and Dilcher 2000). Because most of the species in the Eocene have been reported consistently only in South America, the increase in diversity in the Eocene was mainly driven by in situ origination rather than by immigration from other latitudes (Jaramillo et al., 2010). Our diversity curve confirms the results of Rull (1999), Jaramillo and Dilcher (2000), and Jaramillo et al. (2010) in the sense that the Paleocene exhibits lower diversity than the early Eocene (Fig. 2).

The EECO is the culmination of a long period of global warming and represents the highest ocean temperatures of the last 66 Myr (Miller et al. 1987). An increase in plant diversity has been documented in both southern and northern middle to high latitudes (Christophel 1995; Wing et al. 1995). In the Neotropics, no palynological work has specifically focused on this interval. Our diversity curve shows an outstanding peak at the EECO, which is in fact the second most prominent of the Cenozoic (Fig. 2). Although, the global warming during the EECO could explain the increase in diversity, it is also possible that changes in the paleogeography have played an important role in controlling diversity. The paleogeographic reconstructions by Reyes-Harker et al. (2015) suggest an increase of the area of active deposition between the late Paleocene and the early-middle Eocene. A change from coastal plain and fluvial environments during the late Paleocene to exclusively fluvial environments during the early-middle Eocene is also evident (Reves-Harker et al. 2015). An increase of the geographic area has been cited as an important factor that can boost diversity via the area-species relationship (Rosenzweig 1995). This relationship proposes that larger regions can support more species, which enhance both regional and local diversity by reducing the risk of extinction and increasing niche opportunities (Fine et al. 2011). The predominance of fluvial environments during the early-middle Eocene may have created geographic barriers that promoted diversification through allopatric speciation.

Using the SQS method, we also find evidence for an interval of relatively low plant diversity during the early-middle Eocene, spanning approximately 50-43 Ma, and potentially related to abrupt temperature decreases following the EECO (Fig. 2). In contrast, according to the boundary-crosser diversity estimates using RTM (Supplementary Material 3, Supplementary Fig. 1; see also Jaramillo et al. 2006), diversity remained high during this interval. This discrepancy suggests that taxa that were extirpated from the study region during this interval were not globally extinct, but underwent range shifts resulting in temporary absence or reduction of their abundance, which makes them difficult to detect. This finding, and our broader finding of high volatility of regional extinction and origination during the Paleocene-Eocene, may be consistent with studies using large phylogenies of extant plant species, which find that the Paleocene and Eocene correspond to the periods of the Cenozoic with the highest rates of emigration from the Neotropics (Romero 1993; Wilf et al. 2005; Antonelli et al. 2015). Our findings suggest, therefore, that a shift in the species range out of the Neotropics during the early Eocene was triggered and facilitated by the geographic expansion of the Neotropics during the EECO (Jaramillo and Cárdenas 2013) and the apparent absence of significant physical barriers such as marine corridors or lacustrine systems (Reyes-Harker et al. 2015).

At the Eocene/Oligocene boundary, the planet changed from a warm ice-free greenhouse world to a glacial Antarctic icehouse world (Francis et al. 2008; Westerhold et al. 2020). This transition exhibits a prominent decrease in diversity, which is evident from the upper segment of the middle Eocene, alongside somewhat elevated turnover rates induced by locally high extinction and origination rates (Fig. 5). The declining trend of diversity shows a good correspondence with the cooling trend suggested by the oxygen isotopic curve (Fig. 2). Analysis based on palynology and macrofossils indicates an increase in seasonality in the middle to late Eocene, which changed the plant composition and allowed the establishment of a tropical dry forest (Martínez et al. 2021). Our cluster analysis seems to corroborate the change in plant composition between the middle and late Eocene (Fig. 3). The onset of a

different type of ecosystem should increase ecological heterogeneity and species diversity at regional scale, unless all regional ecosystems are fully replaced by this low-diversity ecosystem. However, the geographic distribution of this new type of ecosystem is still unclear. The work by Martínez et al (2021) is based on a single locality in the Middle Magdalena Valley of Colombia, and to our knowledge there are no data to test whether a tropical dry forest was also present in the Llanos basin at that time. If the establishment of the tropical dry forest was a geographically restricted event, we hypothesize that the

was mainly controlled by the cooling of the planet. Fluctuations in global temperature during the Oligocene to Pliocene were lower in amplitude than those during the Paleocene to Eocene (Fig. 2). In the Neotropics, this change was accompanied by important regional events like the increase of the orogenic activity in the Andean Cordillera (Mora et al. 2010, 2020), the onset of the Amazon River (Hoorn et al. 2017), and the occurrence of regional marine flooding events (Hoorn et al. 2010; Reyes-Harker et al. 2015; Jaramillo et al. 2017; de la Parra et al. 2019). Our results suggest that for diversity, and especially for origination, extinction, and diversification, the Oligocene–Pliocene interval shows relatively less variation than other intervals (e.g., Paleocene to middle Eocene), lacking major extinction pulses but with notable episodes of turnover during the middle Miocene and the Pliocene.

declining trend in diversity during the middle Eocene to Oligocene

The most intense peaks of orogenic activity in the north of the Andean Cordillera occurred during the middle Miocene to early Pliocene (12.4–4.4 Ma) (Hoorn et al. 2010; Mora et al. 2010), which also show a declining trend in global temperature starting at the end of the MMCO and ending in the early Pliocene (Fig. 2). Our diversity curve shows a declining trend from the end of the middle Miocene to the late Miocene and two outstanding peaks in the Pliocene and Pleistocene that are accompanied by relatively elevated volatility in the extinction and origination rates (Fig. 2). It is well established that the intense pulse of orogenic activity during the Miocene-Pliocene promoted allopatric speciation that may have increased the plant diversity in the region. The process of diversification does not exclusively correspond to in situ diversification, but immigrant lineages from both the north (Sklenář et al. 2011) and south (Bacon et al. 2018) played an important role in the historic assembly of the Neotropical flora (Hoorn and Wesselingh 2010). The recent and rapid diversification of the Andes lineages could be explained by massive increase in the area of suitable habitats and opportunities for allopatric speciation, as a consequence of the Andean uplift and late Neogene climate changes (Mora et al. 2008; Bacon et al. 2018).

The increase in diversity since the late Miocene would reflect diversification in the highlands rather than in the lowlands. Our diversity curve captures this diversification, because the composite section integrates information from different sites and therefore from different biomes. It is very likely that this process began in the late Miocene, because the cluster analysis shows that the late Miocene is more similar, in terms of plant composition, to the Pliocene and Pleistocene than to the early or middle Miocene (Fig. 3). The low diversity detected during the middle to late Miocene (Fig. 2) may be the result of a bias in preservation produced by the reconfiguration of the drainage pattern. The influx of sediments from the Andean Cordillera produced alluvial deposits and scarce fine-grained deposits, which are those preserving palynomorphs. In fact, the Guayabo Formation and its lithostratigraphic equivalents (middle to late Miocene) correspond to coarse-grain sandstones, conglomerates, and sporadic claystones

deposited in fluvial–alluvial to deltaic environments (Bayona et al. 2008) in which it is very difficult to recover palynomorphs.

Our analyses have broad implications for understanding interactions between climate and regional extinction and origination. Extinction rates are correlated with directional long-term climate excursions on a timescale of 3 Myr. Origination rates are correlated with extinction in the preceding interval at a timescale of 1 Myr, but the reverse is not true (Table 1). This suggests that global climate state changes have a strong influence on regional extinction and that intervals showing elevated extinction rates are followed by intervals of high origination rates, resulting in an overall pattern of climate-induced turnover. Punctuated climatic events like the EECO, MECO, Oi-1, and MMCO show good correspondence with elevated turnover rates (Fig. 2). These results agree with the idea that the Neotropics have acted as an engine for global plant diversity, with species being formed and replacing each other at unparalleled rates (Antonelli et al. 2015). This engine seems to have run at higher power during the Paleocene and Eocene, and also during the Pliocene. In this sense, the dynamics of Neotropical vegetation during the Cenozoic are comparable to fern dynamics diversity, which is primarily driven by environmentally induced extinctions, with origination being an opportunistic response to diminishing ecospace occupancy (Lehtonen et al. 2017).

Our results shows that there is no straightforward relationship between temperature and Neotropical plant diversity, but climate shifts seem to have had important effects on the dynamics of diversification, with large temperature excursions causing elevated extinction rates. This acts as a mechanism of turnover, because origination is caused by extinction on a timescale of approximately 1 Myr. However, net diversification is approximately zero on intermediate timescales, so the association of standing diversity with climate variation is more complex than a simple association of high temperature with high diversity. Paleogeography also played an important role in controlling plant diversity. The uplift of the Andean Cordillera, the changes in the area of active deposition, and the evolution of the landscape promoted diversification. However, the most relevant paleogeographic changes show a close correspondence with global climate events. Hence, it is difficult to separate the effects of each one, and it is possible that a combination of climate and paleogeography are responsible for modeling plant diversification during the Cenozoic in the Neotropics.

Acknowledgments. We thank Ecopetrol S.A. and especially the Instituto Colombiano del Petróleo (ICP) for the financial and technical support for this work. Special thanks to the Biostratigraphy Team of Ecopetrol and the people of the Department of Earth Sciences of the University of Oxford. We also would like to thank several anonymous reviewers for their useful comments that greatly improved this article and extend special thanks to C. Strömberg and K. Boyce. This research was funded by internal resources of Ecopetrol S.A and it is part of the Ph.D. project of F.d.I.P.

**Competing Interests.** The authors declare no competing interests.

Data Availability Statement. Supplementary Material: Data available through the Zenodo Digital Repository: https://doi.org/10.5281/zenodo.14188005. Supplementary Files: Data available through the Dryad Digital Repository: https://doi.org/10.5061/dryad.rbnzs7hn9.

#### **Literature Cited**

- Alroy, J. 1996. Constant extinction, constrained diversification, and uncoordinated stasis in North American mammals. *Palaeogeography, Palaeoclimatol*ogy, *Palaeoecology* 127:285–311.
- Alroy, J. 2000. Successive approximations of diversity curves: ten more years in the library. *Geology* 28:1023–1026.

- Alroy, J. 2008. Phanerozoic trends in the global diversity of marine invertebrates. *Science* 321:97–100.
- Alroy, J. 2009. Speciation and extinction in the fossil record of North American mammals. Pp. 301–323 in R. Butlin, J. Bridle, and D. Schulter, eds. Speciation and patterns of diversity. Cambridge University Press/British Ecological Society, Cambridge.
- Alroy, J. 2010a. Fair sampling of taxonomic richness and unbiased estimation of origination and extinction rates. *Paleontological Society Papers* 16:55–80.
- Alroy, J. 2010b. The shifting balance of diversity among major marine animal groups. *Science* 329:1191–1194.
- Alroy, J. 2014. Accurate and precise estimates of origination and extinction rates. *Paleobiology* 40:374–397.
- Alroy, J. 2015. A more precise speciation and extinction rate estimator. *Paleo-biology* 41:633–639.
- Alroy, J., R. Marshall, K. Bambach, K. Bezusko, M. Foote, M. Fursich, T. Hansen, et al. 2001. Effects of sampling standardization on estimates of Phanerozoic marine diversification. *Proceedings of the National Academy* of Sciences USA 98:6261–6266.
- Antonelli, A., A. Zizka, D. Silvestro, R. Scharn, B. Cascajales-Minana, and C. Bacon. 2015. An engine for global plant diversity: highest evolutionary turnover and emigration in the American tropics. *Frontiers in Genetics* 6:1–14.
- Bacon, C., F. Velásquez-Puentes, F. Hinojosa, T. Schwartz, B. Oxelman, B. Pfeil, and A. Antonelli. 2018. Evolutionary persistence in Gunnera and the contribution of southern plant groups to the tropical Andes biodiversity hotspot. *PeerJ* 6:e4388.
- Bande, A., B. Horton, J. Ramirez, A. Mora, M. Parra, and D. Stockli. 2012. Clastic deposition, provenance, and sequence of Andean thrusting in the frontal Eastern Cordillera and Llanos foreland basin of Colombia. *Geological Society of America Bulletin* 124:59–76.
- Bayona, G., C. Jaramillo, M. Rueda, A. Reyes-Harker, and V. Torres. 2007. Paleocene–Middle Miocene flexural-margin migration of the nonmarine Llanos Foreland Basin of Colombia. CT&F Ciencia, Tecnología y Futuro 3(3):1–15.
- Bayona, G., A. Valencia, A. Mora, M. Rueda, J. Ortiz, and O. Montenegro. 2008. Stratigraphy and provenance of Miocene rocks in the distal Llanos basin of Colombia. *Geología Colombiana* 33:23–46.
- Benson, R., R. Butler, J. Alroy, P. Mannion, M. Carrano, and G. Lloyd. 2016. Near-stasis in the long-term diversification of Mesozoic tetrapods. *PLoS Biology* 14(1):e1002359.
- Benton, M. 1995. Diversification and extinction in the history of life. *Science* 268:52–58.
- Birks, H., V. Felde, A. Bjune, J. Grytnes, H. Seppä, and T. Giesecke. 2016. Does pollen-assemblage richness reflect floristic richness? A review of recent developments and future challenges. *Review of Palaeobotany and Palynology* 228:1–25.
- Birks, H., and A. Gordon. 1985. Numerical methods in Quaternary pollen analysis. Academic Press, London.
- Bush, A., M. Markey, and C. Marshall. 2004. Removing bias from diversity curves: the effect of spatially organized biodiversity on sampling standardization. *Paleobiology* **30**:666–686.
- Carvalho, M., C. Jaramillo, F. de la Parra, D. Caballero-Rodríguez, F. Herrera, S. Wing, C. D'Apolito, et al. 2021. Extinction at the end-Cretaceous and the origin of modern Neotropical rainforests. *Science* 372:63–68.
- Chao, A., and L. Jost. 2012. Coverage-based rarefaction and extrapolation: standardizing samples by completeness rather than size. *Ecology* 93: 2533–2547.
- Chao, A., C. Chiu, and J. Jost. 2014. Unifying species diversity, phylogenetic diversity, functional diversity, and related similarity/differentiation measures through Hill numbers. *Annual Reviews of Ecology, Evolution, and Systematics* 45:297–324.
- Christophel, D. 1995. The impact of climatic changes on the development of the Australian flora. National Academies Press, Washington, D.C., pp. 174–183.
- Close, R., S. Evers, J. Alroy, and R. Butler. 2018. How should we estimate diversity in the fossil record? Testing richness estimators using samplingstandardized discovery curves. *Methods in Ecology and Evolution* 9(6): 1386–1400.
- Cooper, M., F. Addison, R. Alvarez, A. Hayward, S. Howe, A. Pulham, and A. Taborda. 1995. Basin development and tectonic history of the Llanos Basin,

Eastern Cordillera, and Middle Magdalena Valley, Colombia. AAPG Bulletin 79:1421-1442.

- D'Apolito, C. 2016. Landscape evolution in western Amazonia: palynostratigraphy, palaeoenvironments and diversity of the Miocene Solimoes Formation, Brazil. Ph.D. thesis. University of Birmingham., Birmingham, U.K.
- De la Parra, F., D. Pinzón, G. Rodríguez, O. Bedoya, and R. Benson. 2019. Lacustrine systems in the early Miocene of northern South America, evidence from the upper Magdalena Valley, Colombia. Palaios 34:490-505.
- De la Parra, F., C. Jaramillo, P. Kaskes, S. A. Goderis, P. Claeys, V. Villasante-Marcos, G. Bayona, K. Hatsukawa, and D. Caballero. 2022. Unraveling the record of a tropical continental Cretaceous-Paleogene boundary in northern Colombia, South America. Journal of South America Earth Sciences 114:1-12.
- Dickens, G., J. O'Neil, D. Rea, and R. Owen. 1995. Dissociation of oceanic methane hydrate as a cause of the carbon isotope excursion at the end of the Paleocene. Paleoceanography 10:965-971.
- Edwards, L. 1984. Insights on why graphic correlation (Shaw's method) works. Journal of Geology 92:583-597.
- Felde, V., S. Peglar, A. Bjune, A. Grytnes, and H. Birks. 2016. Modern pollenplant richness and diversity relationships exist along a vegetational gradient in southern Norway. The Holocene 26:163-175.
- Fine, P., R. Ree, and R. Burnham. 2011. The disparity in tree species richness between tropical, temperate and boreal biomes: the geographic area and age hypothesis. Pp. 31-45 in W. Carson and S. Schnitzer, eds. Tropical Forest Community Ecology. Wiley, Oxford, U.K.
- Flannery-Sutherland, J., D. Silvestro, and M. Benton. 2022. Global diversity dynamics in the fossil record are regionally heterogeneous. Nature Communications 13:1-17.
- Foote, M. 1988. Survivorship analysis of Cambrian and Ordovician trilobites. Paleobiology 14:258-271.
- Foote, M. 2000. Origination and extinction components of taxonomic diversity: general problems. Paleobiology 26:74-102.
- Foote, M., and A. Miller. 2006. Principles of paleontology, 3rd ed. Freeman, New York.
- Fraass, A., D. Kelly, and E. Peters. 2015. Macroevolutionary history of the planktic foraminifera. Annual Review of Earth and Planetary Sciences 43: 139-166.
- Francis, J. E., S. Marenssi, R. Levy, M. Hambrey, V. C. Thorn., B. Mohr, H. Brinkhuis, et al. 2008. From greenhouse to icehouse-the Eocene/Oligocene in Antarctica. Developments in Earth and Environmental Sciences 8:309-368.
- Gentry, A. 1982. Neotropical floristic diversity: phytogeographical connections between Central and South America, Pleistocene climatic fluctuations, or an accident of the Andean orogeny? Annals of the Missouri Botanical Garden 69: 557-593.
- Gentry, A. 1992. Tropical forest biodiversity: distributional patterns and their conservational significance. Oikos 63:19-28.
- Gosling, W., F. Mayle, N. Tate, and T. Killeen. 2009. Differentiation between Neotropical rainforest, dry forest, and savannah ecosystems by their modern pollen spectra and implications for the fossil pollen record. Review of Palaeobotany and Palynology 153(1-2):70-85.
- Gosling, W., A. Julier, S. Adu-Bredu, G. Djagbletey, W. Fraser, P. Jardine, B. Lomax, et al. 2017. Pollen-vegetation richness and diversity relationship in the tropics. Vegetation History and Archaeobotany 27:411-418.
- Gregory-Wodzicki, K. 2000. Uplift history of the Central and Northern Andes: a review. Geological Society of America Bulletin 112:1091-1105.
- Gutjahr, M., A. Ridgwell, P. Sexton, E. Anagnostou, P. Pearson, H. Pälike, and G. Foster. 2017. Very large release of mostly volcanic carbon during the Paleocene–Eocene Thermal Maximum. Nature 548:573–577.
- Haffer, J. 1969. Speciation in Amazonian forest birds. Science 165:131-137.
- Hoorn, C. 1993. Marine incursions and the influence of Andean tectonics on the Miocene depositional history of northwestern Amazonia: results of a palynostratigraphic study. Palaeogeography, Palaeoclimatology, Palaeoecology 105:267-309.
- Hoorn, C., and F. Wesselingh. 2010. Amazonia: landscape and species evolution: a look into the past. Wiley, Oxford, U.K.
- Hoorn, C., F. Wesselingh, H. ter Steege, M. Bermudez, A. Mora, J. Sevink, I. Sanmartin, et al. 2010. Amazonia through time: Andean uplift, climate change, landscape evolution, and biodiversity. Science 330:927-931.
- Hoorn, C., G. Bogotá, M. Romero-Baez, E. Lammertsma, S. Flantua, E. Dantas, R. Dino, D. do Carmo, and F. Chemale. 2017. The Amazon at

https://doi.org/10.1017/pab.2024.62

sea: onset and stages of the Amazon River from a marine record, with special reference to Neogene plant turnover in the drainage basin. Global and Planetary Change 153:51-65.

- Hsieh, T., K. Ma, and A. Chao. 2016. iNEXT: an R package for rarefaction and extrapolation of species diversity (Hill numbers). Methods in Ecology and Evolution 7:1451-1456.
- Huber, M., and E. Thomas. 2009. Paleoceanography: the greenhouse world. Pp. 378-389 in J. Steele, S. Thorpe, and K. Turekian, eds. Climate and oceans: a derivative of Encyclopedia of Ocean Sciences, 2nd ed. Academic Press, London, U.K.
- Jablonski, D., K. Roy, and J. Valentine. 2006. Out of the tropics: evolutionary dynamics of the latitudinal diversity gradient. Science 314:102-106.
- Jantz, N., J. Homeier, S. León-Yánez, A. Moscoso, and H. Behling. 2013. Trapping pollen in the tropics-comparing modern pollen rain spectra of different pollen traps and surface samples across Andean vegetation zones. Review of Palaeobotany and Palynology 193:57-69.
- Jaramillo, C. 2019. 140 million years of tropical biome evolution. In J. Gomez and A. Pinilla-Chacon, eds. The geology of Colombia. Servicio Geológico Colombiano, Bogotá. https://doi.org/10.32685/pub.esp.36.2019.06.
- Jaramillo, C., and A. Cárdenas. 2013. Global warming and Neotropical rainforests: a historical perspective. Annual Review of Earth and Planetary Sciences 41:41-766.
- Jaramillo, C., and D. Dilcher. 2000. Microfloral diversity patterns of the late Paleocene-Eocene interval in Colombia, northern South America. Geology 28: 815-818.
- Jaramillo, C., and D. Dilcher. 2001. Middle Paleogene palynology of central Colombia, South America: a study of pollen and spores from tropical latitudes. Palaeontographica, Abteilung B 258:87-213.
- Jaramillo, C., M. Rueda, and G. Mora. 2006. Cenozoic plant diversity in the neotropics. Science 311(5769):1893-1896.
- Jaramillo, C., D. Ochoa, L. Contreras, M. Pagani, H. Carvajal-Ortiz, L. Pratt, S. Krishnan, et al. 2010. Effects of rapid global warming at the Paleocene-Eocene boundary on neotropical vegetation. Science 330:957-961.
- Jaramillo, C., M. Rueda, and V. Torres. 2011. A palynological zonation for the Cenozoic of the Llanos and Llanos Foothills of Colombia. Palynology 35: 46-84.
- Jaramillo, C., I. Romero, C. D'Apolito, G. Bayona, E. Duarte, S. Louwye, J. Escobar, et al. 2017. Miocene flooding events of western Amazonia. Science Advances 3(5):e1601693.
- Juggins, S. 2023. rioja: analysis of Quaternary science data, R package version 1.0-6. https://cran.r-project.org/package=rioja.
- Julier, A., P. Jardine, S. Adu-Bredu, A. Coe, A. Duah-Gyamfi, W. Fraser, B. Lomax, et al. 2017. The modern pollen-vegetation relationships of a tropical forest-savannah mosaic landscape, Ghana, West Africa. Palynology 42: 324-338.
- Kowalewski, M., and P. Novack-Gottshall. 2010. Resampling methods in paleontology. Palentological Society Papers 16:19-54.
- Lazarus, D., J. Barron, J. Renaudie, P. Diver, and A. Turke. 2014. Cenozoic planktonic marine diatom diversity and correlation to climate change. PLoS ONE 9(1):e84857.
- Lehtonen, S., D. Silvestro, D. Karger, C. Scotese, H. Tuomisto, M. Kessler, C. Pena, N. Wahlberg, and A. Antonelli. 2017. Environmentally driven extinction and opportunistic origination explain fern diversification patterns. Scientific Reports 7:4831.
- Leite, F., J. Paz, D. do Carmo, and S. Silva-Caminha. 2016. The effects of the inception of Amazonian transcontinental drainage during the Neogene on the landscape and vegetation of the Solimoes Basin, Brazil. Palynology 41: 412-422.
- Lloyd, G., A. Smith, and J. Young. 2011. Quantifying the deep-sea rock and fossil record bias using coccolithophores. In J. McGowan and A. Smith, eds. Comparing the geological and fossil record. Geological Society of London Special Publication 358:167-177.
- Martínez, C., C. Jaramillo, J. Martínez-Murcia, W. Crepet, A. Cárdenas, J. Escobar, and D. Caballero-Rodríguez. 2021. Paleoclimatic and paleoecological reconstruction of a middle to late Eocene South American tropical dry forest. Global and Planetary Change 205:103617.
- Matthias, I., M. Semmler, T. Giesecke, and M. McGlone. 2015. Pollen diversity captures landscape structure and diversity. Journal of Ecology 103:880-890.

- Miller, K., R. Fairbanks, and G. Mountain. 1987. Tertiary oxygen isotope synthesis, sea level history, and continental margin erosion. *Paleoceanogra*phy 2:1–19.
- Mora, A., M. Parra, M. Strecker, and E. Sobel. 2008. Climatic forcing of asymmetric orogenic evolution in the Eastern Cordillera of Colombia. GSA Bulletin 120:930–949.
- Mora, A., P. Baby, M. Rodazz, M. Parra, S. Brusset, W. Hemorza, and N. Espurt. 2010. Tectonic history of the Andes and sun-Andean zones: implications for the development of the Amazon drainage basin. Pp. 39–59 in C. Hoorn and F. Wesselingh, eds. Amazonia: landscape and species evolution: a look into the past. Wiley, Oxford, U.K.
- Mora, A., D. Villagómez, M. Parra, V. Caballero, R. Spikings, B. Horton, and J. Arias-Martínez. 2020. Late Cretaceous to Cenozoic uplift of the northern Andes: paleogeographic implications. *Geology of Colombia* 3(4):4–121.
- Morrone, J. 2014. Biogeographical regionalisation of the Neotropical region. *Zootaxa* 3782:1–110.
- Parra, M., A. Mora, C. Jaramillo, M. Strecker, E. Sobel, L. Quiroz, M. Rueda, and V. Torres. 2009. Orogenic wedge advance in the northern Andes: evidence from the Oligocene–Miocene sedimentary record of the Medina Basin, Eastern Cordillera, Colombia. *Geological Society of America Bulletin* 121:780–800.
- Perez-Escobar, O., A. Zizka, M. Bermudez, A. Meseguer, F. Condamine, C. Hoorn, H. Hooghiemstra, et al. 2022. The Andes through time: evolution and distribution of Andean floras. *Trends Plant Sciences* 27:364–378.
- Quental, T., and C. Marshall. 2010. Diversity dynamics: molecular phylogenies need the fossil record. *Trends in Ecology and Evolution* **25**:434–441.
- Raup, D. 1972. Taxonomic diversity during the Phanerozoic: the increase in the number of marine species since the Paleozoic may be more apparent than real. *Science* 177:1065–1071.
- R Core Team. 2024. R: a language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. https://www.R-project.org.
- Reyes-Harker, A., C. Ruiz-Valdivieso, A. Mora, J. Ramirez-Arias, G. Rodriguez, F. de la Parra, V. Caballero, et al. 2015. Cenozoic paleogeography of the Andean foreland and retroarc hinterland of Colombia. AAPG Bulletin 99: 1407–1453.
- Roddaz, M., W. Hermoza, A. Mora, P. Baby, M. Parra, F. Christophoul, S. Brusset, and N. Espurt. 2010. Cenozoic sedimentary evolution of the Amazonian foreland basin system. Pp. 61–68 in C. Hoorn and F. Wesselingh, eds. Amazonia, landscape and species evolution: a look into the past. Wiley, Oxford, U.K.
- Romero, E. 1993. South American paleofloras. Biological relationships between Africa and South America. Yale University Press, New Haven, Conn.
- Rosenzweig, M. 1995. Species diversity in space and time. Cambridge University Press, Cambridge.
- Rull, V. 1999. Palaeofloristic and palaeovegetational changes across the Paleocene/Eocene boundary in northern South America. *Review of Palaeobotany* and Palynology 107:83–95.
- Sepkoski, J. 1996. Biodiversity: past, present, and future. *Journal of Paleontology* 71:533–539
- Shaw, A. 1964. Time in stratigraphy, Vol. 1. McGraw-Hill, New York.

- Signor, P., and J. Lipps. 1982. Sampling bias, gradual extinction patterns and catastrophes in the fossil record. *In L. Silver and P. Schultz, eds. Geological implications of impacts of large asteroids and comets on the Earth.* Geological Society of America Special Publication 190:219–296.
- Silva-Caminha, S., C. Jaramillo, and M. Absy. 2009. Neogene palynology of the Solimoes Basin, Brazilian Amazonia. *Paleontographica: Abteilung B* 283(1–3):1–67.
- Silvestro, D., B. Cascales-Minana, C. Bacon, and A. Antonelli. 2015. Revisiting the origin and diversification of vascular plants through a comprehensive Bayesian analysis of the fossil record. *New Phytologist* 207: 425–436.
- Silvestro, D., A. Zizka, C. Bacon, B. Cascales-Minana, N. Salamin, and A. Antonelli. 2016. Fossil biogeography: a new model to infer dispersal, extinction and sampling from palaeontological data. *Philosophical Transactions of the Royal Society B* 371(1691). https://doi.org/10.1098/ rstb.2015.0225.
- Silvestro, D., R. Warnock, A. Gavryushinka, and T. Stadler. 2018. Closing the gap between palaentological and neontological speciation and extinction rate estimates. *Nature Communications* 9:1–14.
- Silvestro, D., N. Salamin, A. Antonelli, and X. Meyer. 2019. Improved estimation of macroevolutionary rates from fossil data using a Bayesian framework. *Paleobiology* 45:546–570.
- Siravo, G., C. Faccenna, M. Gérault, T. W. Becker, M. G. Fellin, F. Herman, and P. Molin. 2019. Slab flattening and the rise of the Eastern Cordillera, Colombia. *Earth and Planetary Science Letters* 512:100–110.
- Sklenář, P., E. Dušková, and H. Balslev. 2011. Tropical and temperate: evolutionary history of paramo flora. *Botanical Review* 77:71–108.
- Stebbins, G. L. 1974. Flowering plants: evolution above the species level. Harvard University Press, Cambridge, Mass.
- Traverse, A. 2007. Paleopalynology. Springer Science & Business Media, New York.
- Weng, C., H. Hooghiemstra, and J. Duivenvoorden. 2006. Response of pollen diversity to the climate-driven altitudinal shift of vegetation in the Colombian Andes. *Philosophical Transactions of the Royal Society B* 362: 253–262.
- Westerhold, T., N. Marwan, A. Drury, D. Liebrand, C. Agnini, E. Anagnostou, and J. Zachos. 2020. An astronomically dated record of Earth's climate and its predictability over the last 66 million years. *Science* 369: 1383–1387.
- Wilf, P., K. Johnson, N. Cuneo, M. Smith, B. Singer, and M. Gandolfo. 2005. Eocene plant diversity at Laguna del hunco and Río pichileufú, Patagonia, Argentina. American Naturalist 165:634–650.
- Wing, S., J. Alroy, and L. Hickey. 1995. Plant and mammal diversity in the Paleocene to early Eocene of the Bighorn Basin. *Palaeogeography, Palaeocli*matology, *Palaeoecology* 115:117–155.
- Wing, S., F. Herrera, C. Jaramillo, C. Gómez-Navarro, P. Wilf, and C. Labandeira. 2009. Late Paleocene fossils from the Cerrejon Formation, Colombia, are the earliest record of Neotropical rainforest. *Proceedings of the National Academy of Sciences USA* 196:18627–18632.